Author’s response to reviews

Title: Community-based participatory research and integrated knowledge translation: advancing the co-creation of knowledge

Authors:

Janet Jull (jjull013@uottawa.ca)
Audrey Giles (agiles@uottawa.ca)
Ian Graham (igraham@ohri.ca)

Version: 3 Date: 01 Dec 2017

Author’s response to reviews:

Dear Dr. Wilson,

Thank you for the response to our most recent re-submission of the paper "Community-based participatory research and integrated knowledge translation: advancing the co-creation of knowledge" (IMPS-D-17-00382R2). It is great news that the resubmitted paper has met editorial approval as potentially acceptable for publication.

The suggested revisions have been made in the text of the abstract: In two instances where it is stated 'CBPR and IKT can both contribute evidence to implementation science' the word 'can' has been substituted for with the words 'have the potential'.

Please pass on my thanks to the editorial team for their dedication and support in working with us on this paper.

With best wishes,

Janet

On behalf of the author team